Author’s response to reviews

Title: Orbital radiotherapy plus three-wall orbital decompression in a patient with rare ocular manifestations of thyroid eye disease: case report

Authors:
Shuo Zhang (akaseki@126.com)
Yang Wang (ocean_wy@foxmail.com)
Sisi Zhong (sisizhong916@163.com)
Xingtong Liu (liuxingtong124@163.com)
Yazhuo Huang (colourful4210@foxmail.com)
Sijie Fang (fangsijie89@hotmail.com)
Ai Zhuang (aizh9h@163.com)
Yinwei Li (dr_yinwei_li@foxmail.com)
Jing Sun (sophiasj@126.com)
Huifang Zhou (fangzzfang@163.com)
Xianqun Fan (fanxq@sjtu.edu.cn)

Version: 2 Date: 20 Jan 2018

Author’s response to reviews:

Dear Professor Francesca Dickens,

We are grateful for your encouraging remarks regarding our manuscript entitled: “Orbital radiotherapy plus three-wall orbital decompression in a patient with rare ocular manifestations of thyroid eye disease: case report”.

According to the remarks, we modified our manuscript including: (1) we have represented authors' names using their full initials in the Authors’ Contributions section and differentiated the
duplicated initials (SZ and SSZ). (2) we have uploaded our manuscript, table and figures as clean versions without any track changes.

We hope that the revised version of the manuscript is now suitable for publication, and thank you in advance for allowing us to contribute to your journal. Please feel free to contact me with any questions regarding the manuscript.

Yours sincerely,
Xianqun Fan
M.D., Ph.D. Professor of Ophthalmology
Department of Ophthalmology, Ninth People's Hospital
Shanghai Jiao Tong University, School of Medicine
No. 639 ZhiZaoJu Road, Shanghai (200011), China
Email: fanxq@sh163.net